Author's response to reviews

Title: Depression and loneliness in Jamaicans with sickle cell disease

Authors:

Monika R Asnani (monika.parshadasnani@uwimona.edu.jm)
Raphael Fraser (raphael.fraser@uwimona.edu.jm)
Norma A Lewis (norma.lewis@uwimona.edu.jm)
Marvin E Reid (marvin.reid@uwimona.edu.jm)

Version: 3 Date: 3 June 2010

Author's response to reviews: see over
June 3, 2010.

Sabina Alam  
Senior Scientific Editor  
BMC series journals

Dear Dr. Alam,

Please see formatting changes to the manuscript entitled “Depression and Loneliness in Jamaicans with Sickle Cell Disease”.  
We trust that you will find these changes acceptable.

Sincerely Yours,

Dr. Monika R. Asnani,  
Research Fellow,  
Sickle Cell Unit, University of the West Indies, Mona  
Kingston 7, Jamaica  
Tel: (876)927-2471 Fax : (876)927-2984.  
e-mail: monika.parshadashani@uwimona.edu.jm
Major changes

1. Names were formatted as required.
2. Author contributions were rewritten as required. Page 15-16: “Authors’ contributions: MA contributed to study design, data collection, analysis and manuscript preparation; RF contributed to study design and data analysis; NL contributed to data collection and final manuscript preparation; MR contributed to design, data collection, analysis and manuscript preparation. All authors have read and approved the final manuscript.”

Minor changes

1. Key words were removed.
2. All trackings were removed from the manuscript.
3. Table formats were changed as required.
4. The figure was removed from the main manuscript and the figure legend was placed at end of references.
5. The figure title was removed from the image file.
6. The typography was rechecked.